Author’s response to reviews

Title: Glycated Hemoglobin Versus Oral Glucose Tolerance Test in The Identification of Subjects With Prediabetes in Qatari Population

Authors:

Muhammad Abdul-Ghani (abdulghani@uthscsa.edu)
Saadallah Iskandar (IAKANDAR@QP.COM.QA)
Ayman Megahid (YMNDODO20@GMAIL.COM)
Dalia Kamal (DKAMAL@HAMAD.QA)
Osama Migahid (PREDRMIGAHID@GMAIL.COM)
Ralph DeFronzo (ALBARADO@UTHSCSA.EDU)
Amin Jayyousi (AJAYYOUSI@HAMAD.QA)
Mahmood Al Jaidah (ALJAIDAH@QP.COM.QA)

Version: 2 Date: 27 Jun 2019

Author’s response to reviews:

1. We note the removal of one or more authors since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

In such cases, we use a standardised form which we would request that you and your co-authors complete. The change in authorship form can be found from the link on the following page:

https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5

Please also ensure that the author list in the submission is updated to match the list in the manuscript.

We have maintained the original list of authors in the revised manuscript.
2. In the Ethics approval section of the Declarations, please write out the full name of the ethics committee that approved your study and include a reference number if, available.

We have added the name of IRB board, HMC Ethical committee, to the revised manuscript.

3. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript. The statement regarding authors’ approval of the final manuscript should instead be included in the Authors’ contributions section of the Declarations.

We agree to the terms of the consent.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

A clean version of all components of the manuscript have been uploaded.